Author's response to reviews

Title: A case-crossover study of alcohol consumption, meals and the risk of road traffic crashes

Authors:

Stefano Di Bartolomeo (stefano.dibartolomeo@sanita.fvg.it)
Francesca Valent (francesca.valent@uniud.it)
Riccardo Marchetti (paninetti@yahoo.it)
Rodolfo Sbrojavacca (rsbrojav@tin.it)
Fabio Barbone (fabio.barbone@uniud.it)

Version: 4 Date: 27 August 2009

Author's response to reviews:

Table order - please cite all Tables in numerical order in the text. Table 4 is currently cited out of sequence.

First citation of table 4 eliminated, sequence now ok.